**Author's response to reviews**

**Title:** A before and after study of the impact of academic detailing on the use of diagnostic imaging for shoulder complaints in general practice.

**Authors:**

- Norm A Broadhurst (norm.broadhurst@flinders.edu.au)
- Christopher A Barton (christoher.barton@adelaide.edu.au)
- Debra Rowett (Debra.Rowett@rgh.sa.gov.au)
- Lisa Yelland (lisa.yelland@adelaide.edu.au)
- David K Martin (dkmartin@bigpond.net.au)
- Angela Gialamas (angela.gialamas@adelaide.edu.au)
- Justin J Beilby (justin.beilby@adelaide.edu.au)

**Version:** 5  **Date:** 14 March 2007

**Author's response to reviews:** see over
Changes to manuscript arising from Core pre-accept

- Title – changed from upper to all sentence (lower) case.
- Highlight/tracking – removed.
- Author List – forename initial surname
- Country – Country included.
- Introduction – name changed to background
- References – consecutive numbers starting from 1 in square brackets.
- reference placed before full stop
- reference list contains all authors names
- Section headings – Sentence case used
- Competing interests – the statement “the authors declare that they have no competing interests” has been inserted
- Authors contributions – authors’ contributions
- Figure title – removed from figure and included on a new page after the reference list.
- Tables – have been copied and pasted into the main manuscript document
- Additional files – have been added to the main document after the tables and named as requested.